Author's response to reviews

Title: Evaluation of high-risk Human papillomaviruses type distribution in cervical cancer in Sichuan province of China

Authors:

En-qi Wu (w1n2q3@yahoo.com.cn)
Guo-nan Zhang (zhanggn@hotmail.com)
Xiang-hui Yu (xianghui@jlu.edu.cn)
Yuan Ren (renyuan811@163.com)
Ying Fan (fanying51@163.com)
Yong-ge Wu (yonggewu@163.com)
Wei Kong (weikong@jlu.edu.cn)
Xiao Zha (zha530909@sina.com)

Version: 4 Date: 12 July 2008

Author's response to reviews:

Dear Editor:

Thank you very much for your letter regarding our manuscript entitled “Evaluation of high-risk Human papillomaviruses type distribution in cervical cancer in Sichuan province of China.” We have checked the manuscript and revised it according to the comments. We submit here the revised manuscript as well as a list of changes.

Major revisions:

1. We have moved the Abstract section to a new page after the tile page.

2. The references were formatted in the BMC reference style. All reference titles have been written in bold.

3. We added Figure Legend section in the manuscript after the References section.

Minor revisions:

1. The keywords were removed from the Abstract section.

2. We corrected the format of Authors’ contributions section, using initials to refer to each authors’ contribution.
3. We removed the “6” prior to the “References”, reformatted reference 1.

4. All authors’ names were included in the reference list.

5. We removed the number and title of the figure from image files, and put them in the Figure Legend section of the manuscript.

6. All figures were cropped to minimize the amount of white space surrounding the illustration.

Thank you very much in advance for your work.

Sincerely yours,

Xiao Zha

Department of Molecular Pharmacology and Gynecol Oncology,

Sichuan Tumor Hospital, Chengdu 610041, China